
Hall A Report

Hall A: Ciprian Gal - 13th of Dec 2016 -

• Wed-Monday 8AM: 

• efficiently took data 

• DVCS collected 6.7% of 
the total approved days 
(up to 42.9% now)

• Monday: 

• configuration change to 3 pass 

• elastic calibration went smoothly through the night 

• Tuesday: 

• continuing the calibration until 1PM 
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• Week ahead: 

• we are taking a 2-3 hour access for a configuration 
change to 4 pass 

• RF separator work will continue today on MCC side 
however the priority has changed to Hall D and if 
things don’t work as expected they will most likely 
magnetically steer the beam to Hall D sometime this 
evening 

• Run whenever we get beam


